# Experimental evolution of metabolism under nutrient restriction: enhanced amino acid catabolism and a key role of branched-chain amino acids

Fanny Cavigliasso<sup>1, (D)</sup>, Loriane Savary<sup>1</sup>, Jorge E. Spangenberg<sup>2, (D)</sup>, Hector Gallart-Ayala<sup>3, (D)</sup>, Julijana Ivanisevic<sup>3, (D)</sup>, Tadeusz J. Kawecki<sup>1, (D)</sup>

<sup>1</sup>Department of Ecology and Evolution, University of Lausanne, Lausanne, Switzerland <sup>2</sup>Institute of Earth Surface Dynamics, University of Lausanne, Lausanne, Switzerland <sup>3</sup>Metabolomics Unit, Faculty of Biology and Medicine, University of Lausanne, Lausanne, Switzerland

Corresponding authors: Department of Ecology and Evolution, University of Lausanne, Biophore, CH 1015, Lausanne, Switzerland. Email: fanny.cavigliasso@gmail. com; Department of Ecology and Evolution, Biophore, CH 1015, University of Lausanne, Lausanne, Switzerland. Email: fadeusz.kawecki@unil.ch

### Abstract

Periodic food shortage is a common ecological stressor for animals, likely to drive physiological and metabolic adaptations to alleviate its consequences, particularly for juveniles that have no option but to continue to grow and develop despite undernutrition. Here we study changes in metabolism associated with adaptation to nutrient shortage, evolved by replicate *Drosophila melanogaster* populations maintained on a nutrient-poor larval diet for over 240 generations. In a factorial metabolomics experiment we showed that both phenotypic plasticity and genetically-based adaptation to the poor diet involved wide-ranging changes in metabolite abundance; however, the plastic response did not predict the evolutionary change. Compared to nonadapted larvae exposed to the poor diet for the first time, the adapted larvae showed lower levels of multiple free amino acids in their tissues—and yet they grew faster. By quantifying accumulation of the nitrogen stable isotope <sup>15</sup>N we show that adaptation to the poor diet led to an increased use of amino acids for energy generation. This apparent "waste" of scarce amino acids likely results from the trade-off between acquisition of dietary amino acids and carbohydrates observed in these populations. The three branched-chain amino acids (leucine, isoleucine, and valine) showed a unique pattern of depletion in adapted larvae raised on the poor diet. A diet supplementation experiment demonstrated that these amino acids are limiting for growth on the poor diet, suggesting that their low levels resulted from their expeditious use for protein synthesis. These results demonstrate that selection driven by nutrient shortage not only promotes improved acquisition of limiting nutrients, but also has wide-ranging effects on how the nutrients are used. They also show that the abundance of free amino acids in the tissues does not, in general, reflect the nutritional condition and growth potential of an animal.

Keywords: experimental evolution, metabolomics, dietary restriction, deamination, malnutrition, Drosophila melanogaster

### Lay Summary

Juvenile animals are particularly vulnerable to nutrient deprivation—they usually do not have an option of arresting their development and just trying to survive until food becomes plentiful; rather, they must attempt to grow and develop with whatever nutrients that can scrape. While they could obviously improve their lot by getting better at finding the scarce food, could they also adapt their physiology and metabolism in a way that would alleviate consequences of undernutrition? To find out we let populations of the fruit fly *Drosophila melanogaster* adapt during 240 generations to conditions of chronic larval nutrient shortage, and then studied their metabolism. We found that these populations evolved changes in their amino acid metabolism: their larvae are better at extracting amino acids from nutritionally poor diet and are able to grow faster (which requires a higher rate of protein synthesis), while maintaining lower levels of most amino acids in their system. This suggests improved cellular "logistics," with a higher turnover of raw materials associated with their lower stocks owing to their more efficient and immediate use. However, paradoxically, the malnutrition-adapted larvae also "waste" a substantial fraction of their amino acids by "burning" them for energy. They can afford this because of their improved extraction of scarce amino acid from the diet, but they may also be compelled to do this by a trade-off with absorption of dietary carbohydrates.

## Introduction

Many animal populations experience periods of nutrient shortage that result in physiological stress and reduced fitness. If these periods occur repeatedly over generations, they will generate natural selection for physiological adaptations that alleviate their consequences. These adaptations include accumulating metabolic reserves when nutrients are plentiful, and the ability to shut down reproduction and switch to a "survival mode." This mode is characterized by reduced metabolism, mobilization of fat stores, and in extremis, autophagy, when nutrients become scarce (Tatar et al., 2003; Singh et al., 2019). They help the animal wait out famines and survive to reproduce once food becomes plentiful again. However, these strategies are often unavailable to juveniles exposed to prolonged undernutrition. With some exceptions like

Received February 6, 2023; revisions received April 4, 2023; accepted April 24, 2023

<sup>©</sup> The Author(s) 2023. Published by Oxford University Press on behalf of The Society for the Study of Evolution (SSE) and European Society for Evolutionary Biology (ESEN).

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

the *dauer* larva of *Caenorhabditis elegans*, juveniles cannot arrest their development and wait out the lean times; their only option is to attempt to develop and grow using whatever nutrients they can obtain. This is likely to favor adaptations in nutrient acquisition, metabolism, and nutrient use different from those favoring famine survival in adults. Exploring these adaptations may thus help understanding aspects of metabolism and physiology of juvenile animals. Furthermore, given increasing evidence that aspects of metabolism appear programmed for life, it may contribute to explaining aging (de Magalhaes, 2012; Carlsson et al., 2021) and human vulnerability to metabolic disease (Prentice, 2005; Roseboom et al., 2006).

In this study, we explore changes in metabolism associated with genetically-based adaptation to juvenile (larval) undernutrition evolved in a long-term laboratory natural selection experiment with Drosophila melanogaster. The experiment involves six "Selected" populations that evolved for more than 240 generations on a poor (nutritionally diluted) larval diet and six "Control" populations maintained in parallel on a standard (sufficient, but not exceedingly nutrient rich) diet. The poor diet results in a drastic reduction of larval growth, such that nonadapted larvae take more than twice as long to develop (from egg to pupa) than on standard diet, and yet emerge at less than half the weight of flies raised on the standard diet (Kolss et al., 2009; Vijendravarma et al., 2011). In the course of experimental evolution, the Selected populations acquired the ability to grow and develop substantially faster on the poor diet compared to Control populations (Kolss et al., 2009; Erkosar et al., 2017; Cavigliasso et al., 2020). They also evolved a smaller critical size for metamorphosis initiation (Vijendravarma et al., 2012); as a consequence, they emerge as smaller adults than Controls despite their faster growth on poor diet (Vijendravarma et al., 2011; Cavigliasso et al., 2020).

The faster growth of Selected larvae on poor diet appears at least partly mediated by improved acquisition of limiting nutrients. Selected larvae assimilate amino acids from poor diet at a higher rate than Controls, but are less efficient in assimilating sucrose (Cavigliasso et al., 2020), consistent with changes in the expression of digestive enzymes (Erkosar et al., 2017). An adaptive character of this change is supported by the finding that larval growth on the poor diet is limited by yeast (the main source of dietary protein) but not by sugar availability (Cavigliasso et al., 2020). Thus, experimental adaptation of Selected populations to larval undernutrition involved postingestive nutritional compensation, i.e., a shift in investment in the acquisition of the major categories of dietary macronutrients (Behmer, 2009; Simpson et al., 2012).

However, it is unlikely that selection imposed by the dietary regimes was limited to mechanisms of nutrient acquisition. Whole genome sequencing identified signatures of allele frequency differentiation between Control and Selected populations at single nucleotide polymorphisms (SNPs) in over 100 genomic regions (Kawecki et al., 2021). These results point to a highly polygenic nature of adaptation to diet in these populations, suggesting the involvement of many physiological mechanisms. This observation is also supported by a large number of differentially expressed genes with diverse functions (Erkosar et al., 2017). Both candidate SNP-associated and differentially expressed genes include many genes involved in metabolism (notably in lipid and carbohydrate metabolism) and in the regulation of response to nutrients and growth, such as insulin and FOXO signaling (Erkosar et al., 2017; Kawecki et al., 2021). Therefore, the adaptation of Selected populations to the poor diet was mediated not only by changes in the acquisition of scarce nutrients but also by changes in the way the acquired nutrients were used in metabolism and growth (predictions from this article are listed in Supplementary Table S1).

Here we investigate such evolved differences in metabolism between Selected and Control larvae, particularly focusing on amino acids. Under the simplest scenario, the faster assimilation of amino acids from the poor diet by Selected larvae could directly feed into an increased rate of protein synthesis, leading to faster growth without significant changes in other aspects of metabolism. However, in addition to being building blocks of proteins, dietary amino acids are substrates for the synthesis of many important molecules, such as nucleotides, catecholamine hormones, neurotransmitters, and cofactors (Wu et al., 2014), and some play a role in signaling (Manière et al., 2020). Furthermore, amino acids may be deaminated and the carboxylic skeleton may be used to generate ATP or energy stores (i.e., glycogen and fat; Soultoukis & Partridge, 2016; Holtof et al., 2019). This last use of amino acids by larvae faced with nutrient shortage might appear wasteful, especially that protein to carbohydrate ratio in both the standard and the poor diet is lower than optimal for Drosophila larvae (Rodrigues et al., 2015; Cavigliasso et al., 2020). Yet, processes that underlie growth (foraging, digestion transport of nutrients, protein synthesis) require energy. Furthermore, larvae need to accumulate triglyceride reserves to fuel the energetically costly process of metamorphosis (Boggs, 2009). Because of their faster growth on poor diet, the Selected larvae presumably need more energy per time unit than Controls; yet, they are less efficient in assimilating dietary carbohydrates (Cavigliasso et al., 2020). These arguments lead to contrasting predictions: either selection under the poor diet regime may have favored minimizing the use of scarce amino acids for energy generation, or it may have favored their greater use to compensate for lower carbohydrate assimilation and increased energy need.

We first address the metabolic consequences of adaptation to poor diet with high-coverage targeted metabolomics, comparing the patterns of abundance of core metabolites, in particular amino acids and their derivatives and compounds involved in energy metabolism, between Selected and Control larvae. We employ a factorial design, whereby the metabolome of larvae from both Selected and Control populations is assessed on both poor and standard diets. Because the base population from which these populations were derived had been maintained on the standard diet for 5 years before the onset of the evolution experiment, it should have already been adapted to this diet. We therefore interpret metabolic differences between Selected and Control larvae raised on poor diet as a result of evolutionary change of Selected populations driven by the poor diet evolutionary regime. We address caveats of this interpretation in Supplementary Discussion.

Additionally, a comparison of the metabolome of larvae of the same population raised on poor versus standard diet represents a phenotypically plastic response of that population to nutritional conditions. This comparison is of interest in its own right, given the proposed role of free amino acids as indicators of nutrient availability, relied on by TOR pathway to regulate cell growth and division (Antikainen et al., 2017). However, the effect of dietary limitation on the level of amino acids and other metabolites in the larvae does not seem to have been assessed; in adult flies nutritional restriction strongly affects amino acids and studies (Avanesov et al., 2014; Laye et al., 2015; Jin et al., 2020).

We also explore the relationship between phenotypic plasticity and genetically-based adaptation. It is sometimes claimed that evolutionary adaptation to novel environments primarily occurs by natural selection favoring genetic variants that magnify the initial phenotypically plastic responses and eventually lead to their constitutive expression (Pigliucci & Murren, 2003; Paenke et al., 2007; Lande, 2009). However, evidence from gene expression patterns tends to be more consistent with a negative correlation between plastic responses and evolutionary changes (Yampolsky et al., 2012; Ghalambor et al., 2015; Huang & Agrawal, 2016; Josephs et al., 2021). Several authors proposed that a positive (respectively negative) correlation between plasticity and evolution in gene expression patterns implies that plasticity is predominantly adaptive (respectively maladaptive; Ghalambor et al., 2015; Huang & Agrawal, 2016; Josephs et al., 2021). Our results suggest that this interpretation may be too simplistic.

The results of metabolic profiling revealed interesting patterns of differentiation in the abundance of amino acids and purine compounds, leading to two further specific questions that we addressed with additional experiments. First, we used amino acid supplementation to test for the role of branched-chain amino acids in adaptation to the poor diet. Second, we used a stable isotope-based approach to compare the rate of amino acid deamination between Selected and Control larvae. We describe the rationale for these experiments in the Results section.

### Methods

This is an abbreviated outline of the methods; a detailed version is available as Supplementary Methods.

#### **Experimental evolution**

We used six Control and six Selected populations of Drosophila melanogaster originally derived from an outbred base population adapted to the lab and to the standard diet (Kolss et al., 2009). The Control populations have been maintained at a controlled larval density on our standard yeast-cornmeal-sugar larval diet medium (1.25% yeast w/v). The six Selected populations have been maintained on the same density on a poor larval diet containing one-fourth of the amount of nutrients of the standard diet. After their emergence, adult flies in both evolutionary regimes were transferred to standard diet. The experiments described in this article were performed after 244-287 generations of experimental evolution. To obtain larvae for each experiment, a controlled number of eggs (the number depending on experiment, see Supplementary Methods) was transferred onto the experimental food media and inoculated with fly feces to ensure the homogeneity of larval microbiota. Parents and grandparents of the assayed larvae were all raised on standard diet to minimize nongenetic parental effects.

#### Multiple pathway targeted metabolomics

We performed targeted metabolomics on whole early third instar larvae, targeting metabolites involved in multiple core energy pathways (glycolysis, pentose phosphate pathway, amino-acid metabolism, purine and pyrimidine metabolism, TCA cycle and OXOPHOS, beta-oxidation). We implemented a factorial design, with larvae from all six Selected and six Control populations (factor regime) raised on both standard and poor diets (factor diet). We collected three replicate samples from each population on each diet (arranged in batches). Some samples were filtered out, leaving a total of 65, in which we quantified 174 metabolites. Logtransformed metabolite abundance, normalized to larval protein content, was analyzed with a linear mixed model, with diet (poor vs. standard), regime (Selected vs. Control) and their interaction as fixed factors, and the replicate population and batch as random factors.

#### BCAA supplementation experiment

Metabolome results suggested alternative hypotheses about the patterns of abundance of branched-chain amino acids (BCAA) that led to contrasting predictions about the consequences of supplementing the poor diet with those amino acids, as described in Results. To test these predictions, we raised larvae of Selected and Control populations on the poor diet and on the poor diet supplemented with the three BCAA (valine, leucine, and isoleucine) in amounts that resulted in doubling of their concentrations contained in yeast in the diets. To verify that any effect is specific to BCAA and not a consequence of adding any amino acids, we included a third treatment, with the poor diet supplemented with three other essential amino acids, tryptophan, histidine, and lysine, also doubling their concentration (WHK treatment). We recorded the egg-to-adult survival probability, the developmental rate (inverse of developmental time) and female dry weight and the combined developmental time and weight to estimate larval growth rate. This experiment was done in three blocks. To analyze the results, we fitted a generalized linear mixed model (for egg-to-adult survival probability) or linear mixed models (for the other traits) with treatment, regime, and their interaction as fixed factors, and the replicate population and block as random factors.

#### <sup>15</sup>N accumulation experiment

Metabolome data suggested differences in the rate at which amino acids are deaminated and the nitrogen excreted as waste. To verify this, we exploited the fact that the amino acid deamination process has a slight preference for the common light isotope of nitrogen (<sup>14</sup>N). As a consequence, proteins in tissues become enriched in the heavy isotope of nitrogen (<sup>15</sup>N) to a degree that increases with rate of use of amino-acids deamination (Minagawa & Wada, 1984; Peterson & Fry, 1987; Gannes et al., 1997). Because <sup>15</sup>N is naturally rare, in this experiment we used food media containing yeast enriched in <sup>15</sup>N, which improved signal to noise ratio.

We raised larvae from egg on the  $^{15}N$ -enriched standard and poor diet and collected samples of 20–30 prepupae (not sexed) and 20–30 freshly emerged virgin females. Two samples per population, diet, and life stage were collected in two blocks separated in time.  $^{15}N$  content ( $\delta^{15}N$  value in per mil relative to standard air nitrogen, ‰  $N_2$ -Air) was measured in each sample and normalized to  $^{15}N$  in the diet.  $\Delta^{15}N$  ( $\delta^{15}N_{prepupae or adult} - \delta^{15}N_{diet}$ ) was analyzed with a linear mixed model with regime, diet, stage (prepupae vs. adult females) and their interactions as fixed factors, and the replicate population and block as random factors.

### Results

#### Metabolite abundance differs between Selected and Control larvae especially when raised on poor diet

As expected, the diet treatment had a strong impact on metabolite abundances, with principal component analysis (PCA) clearly separating larvae raised on poor and standard diet, mainly along the first PC axis (Figure 1A; Wilks  $\lambda = 0.01$ ,  $F_{3,18} = 625.1$ , p < .001, MANOVA on PC1-3, Supplementary Table S2). In turn, the second and third PCs differentiated the Selected and Control populations (Figure 1A; Wilks  $\lambda = 0.14$ ,  $F_{3,18} = 35.8$ , p < .001, MANOVA on PC1-3, Supplementary Table S2). The Selected larvae were differentiated from Controls in the PC1-3 space irrespective of diet (MANOVA contrasts,  $t_{20} = 8.8$ , p < .001 on poor diet,  $t_{20} = 5.2$ , p < .001 on



**Figure 1.** Diet quality induces both phenotypically plastic responses ("Diet") and evolutionary changes ("Regime") in metabolism. (A) PCA on the abundance of all 174 metabolites; each point corresponds to a population estimate averaged over replicate samples (an animated version is available in Supplementary Figure S1). (B) Venn diagram summarizing the overlaps between significant metabolites. For diet this includes the main effect whereas for regime it includes both the main effect of regime and the regime contrast on poor diet (see text).

standard diet). However, this difference was more pronounced on poor diet as manifested as regime × diet interaction in MANOVA on PC1-3 (Wilks  $\lambda = 0.58$ ,  $F_{3.18} = 4.4$ , p = .018).

The median estimated effect size (absolute  $\log_2$  fold-change) was twice as large for diet than for regime (0.28 vs. 0.14, p < .001, Wilcoxon test). The median absolute effect of regime was smaller in larvae raised on poor than on standard diet (0.12 vs. 0.19, p < .001, Wilcoxon test), but even the latter was still smaller than the effect of diet (0.19 vs. 0.28, p < .001, Wilcoxon test). Thus, the effect of diet on metabolome was generally larger in magnitude than the effect of the evolutionary regime (Supplementary Figure S2).

Of 174 metabolites that passed quality assessment, univariate factorial analysis of abundance identified 113 affected by diet, and 54 that showed a significant effect of regime (i.e., Selected vs. Control populations) at 10% FDR; for 13 metabolites we detected regime × diet interaction. In a diet-specific contrast analysis 53 metabolites differed in abundance at 10% FDR between Control and Selected larvae raised on poor diet but only one differed on standard diet (Supplementary Table S3). In total 68 metabolites were detected as significantly different between Selected and Control lines at 10% FDR for as the main effect in the analysis (i.e., across diets) and/or in larvae raised on the poor diet. In the following we consider these 68 metabolites as significantly affected by evolution driven by diet (Figure 1B).

# Plastic response to poor diet involves accumulation of many metabolites

The response of each population to the diet treatment represents phenotypic plasticity (Scheiner, 1993). Presumably, the response of Control larvae to poor diet approximates the ancestral plastic response of the base population, and thus the starting point for evolutionary adaptation of the Selected populations when they were first exposed to the poor diet. We thus examined the effect of diet on metabolite abundance in Control larvae. We identified 44 (25%) metabolites less abundant and 62 (36%) more abundant on the poor diet, spread among different metabolite groups (Figure 2, Supplementary Table S3). Metabolites with higher abundance on poor diet were particularly numerous among proteinogenic amino acids (11 out of 19), amino acid derivatives containing the amino group (24/50), and vitamins of B group (5/8). Thus, rather unexpectedly, dietary nutrient shortage resulted in an increase in tissue concentrations of many metabolites.

# Adaptation to malnutrition involves changes in amino acid and purine metabolism

The phenotype expressed on the poor diet has been the target of selection in Selected populations. Differences between poor dietraised Selected and Control (second columns of heat maps in Figure 2) thus represent the genetically-based evolutionary change driven by the diet. The 68 metabolites differentially abundant between Selected and Control larvae included chemically and functionally diverse compounds involved in a range of metabolic pathways. Nearly 3/4 of them (49) were less abundant in Selected than Control larvae while only 19 were more abundant, a significant deviation from 50:50 (sign test, p < .001). This includes lower abundance of several metabolites involved in sugar metabolism in Selected larvae, but higher abundance of a few that link glycolysis with triglyceride (2/3-phosphoglycerate, glycerol-3-phosphate) and glycogen (UDPglucoronate) storage. Of the four building blocks of RNA, guanosine and uridine were significantly less abundant while adenosine and cytidine tended to be less abundant. Of 19 proteogenic amino acids, 9 were significantly less abundant in Selected than Control populations while none was more abundant (Figure 2). Multiple amino acid derivatives containing the amino group were likewise affected by evolutionary regime, suggesting differences in amino acid catabolism. We also observed several changes in the purine/uric acid synthesis pathway, notably a higher concentration of AICAR, xanthine and urate in Selected compared to Control larvae. As we argue below, these differences suggest that Selected larvae catabolize a greater fraction of their amino acids, a hypothesis we test with an independent experiment.

# Plasticity of metabolite abundance in response to diet does not predict evolutionary response

As summarized in Introduction, plasticity is often proposed to anticipate evolutionary change driven by the same environmental



| Proteinogenic amino acids           | (poor-std)  | (SEL     | -CTL)                 | • | Sugar         |
|-------------------------------------|---|----------|-----------------------|---|---------------|
| ISOLEUCINE                          | (poor std)  | × (021   | *                     |   | meta          |
| LEUCINE                             | *   | ×        | *                     |   |               |
| VALINE                              |   | ×        | *                     |   |               |
| ALANINE                             |   |          | *                     |   |               |
| PHENYLALANINE                       |   |          | *                     |   |               |
| GLUTAMATE                           |   | ,        | *                     |   |               |
| ARGININE                            | *   | î        | *                     |   |               |
| HISTIDINE                           | •   |          |                       |   |               |
| TRYPTOPHAN                          |   |          |                       |   |               |
| GLYCINE                             |   |          |                       |   |               |
| SERINE                              |   |          |                       |   |               |
| TYROSINE                            | *   |          |                       |   |               |
| ASPARAGINE                          | •   |          |                       |   |               |
| ASPARTATE                           | *   |          |                       |   |               |
| GLUTAMINE                           |   |          |                       |   | IM.           |
| METHONINE                           |   |          |                       |   |               |
| Amino acid derivatives containing N | (near atd)  | (CEI     | CTU                   |   |               |
| netabolite                          | (poor-sta)  | CTL (SEL | -GTL) <sub>poor</sub> |   |               |
| DIMETHYLGLYCINE                     |   |          | *                     |   |               |
| KYNURENATE                          | *   |          |                       |   |               |
| CARNOSINE                           | *   |          | *                     |   |               |
| XANTHURENATE                        | •   | ×        |                       |   | Vites         |
| LADAVERINE                          |   | ×        | *                     |   | vitar<br>mete |
| TAURINE                             | *   |          | *                     |   |               |
| GLUTATHIONE OXIDIZED                |   |          |                       |   |               |
| 1-AMINOCYCLOPROPANECARBOXYLATE      | *   |          | *                     |   |               |
| 2-AMINOISOBUTYRATE                  |   |          | *                     |   |               |
| ETHANOLAMINE                        | •   |          | *                     |   |               |
| 4-GUANIDINOBUTANOATE                |   |          |                       |   |               |
| UROCANATE                           | •   |          |                       |   |               |
| PUTRESCINE                          |   |          |                       |   | -             |
| TRYPTOPHANAMIDE                     |   |          |                       |   | meta          |
| N-FORMYLMETHIONINE                  |   | _        | *                     |   | mete          |
| DIMETHYLARGININE                    |   |          |                       |   |               |
| AMINOADIPATE                        |   |          |                       |   |               |
|                                     | •   |          |                       |   |               |
| HYDROXYKYNURENINE                   | •   | ×        |                       |   |               |
| ANTHRANILATE                        | *   |          |                       |   |               |
| CYSTINE                             | •   |          |                       |   |               |
| BE IA-ALANINE/SARCOSINE             | •   |          |                       |   |               |
| GLUTATHIONE REDUCED                 | *   |          |                       |   |               |
| OXOPROLINE                          | •   |          |                       |   | Acyc          |
| N-ACETYLSERINE                      | *   |          |                       |   | meta          |
| N-METHYLASPARTATE                   |   | _        |                       |   |               |
| CITRULLINE                          |   |          |                       |   |               |
| HISTAMINE                           |   |          |                       |   |               |
| KYNURENINE                          |   |          |                       |   |               |
| 5-AMINOLEVULINATE                   |   |          |                       |   |               |
| S-ADENOSYLMETHIONINE                |   |          |                       |   |               |
| GAMMA-AMINOBUTYRATE                 |   |          | *                     |   |               |
| N-ACETYLPUTRESCINE                  |   | ×        |                       |   |               |
| N-ACETYLLEUCINE                     | *   |          |                       |   |               |
| N-ACETYLASPARTATE                   |   |          | *                     |   |               |
| N-ACETYLGLUTAMATE<br>O-ACETYLSERINE |   |          | *                     |   |               |
| 4-ACETAMIDOBUTANOATE                | *   | ×        | *                     |   | _             |
| CREATINE                            |   |          | *                     |   | Othe          |
| N,N,N-TRIMETHYLLYSINE               | *   |          | *                     |   | meta          |
| 4-HYDROXY-PROLINE                   |   |          |                       |   |               |
| THOTAURINE                          |   |          |                       |   |               |
| Purine compounds                    | (near atc)  | /00      | -CTL)                 |   |               |
| The capolite                        | (poor-std)  |          | *                     | 1 |               |
| GUANINE                             |   |          | *                     |   |               |
| GUANOSINE                           |   | ×        | *                     |   |               |
| INOSINEMONOPHOSPHATE                | *   |          | *                     |   |               |
| FAD                                 |   |          | *                     |   |               |
| INOSINE                             |   |          |                       |   |               |
| AMP                                 |   |          |                       |   |               |
| CYCLIC AMP                          |   |          |                       |   |               |
| ADENOSINE                           | *   |          |                       |   |               |
|                                     |   |          |                       |   |               |
| dGMP                                |   |          |                       |   |               |
| NADP                                |   |          |                       |   |               |
| GMP                                 |   |          |                       |   |               |
| ATP                                 |   |          |                       |   |               |
| XANTHINE                            | *   |          | *                     |   |               |
| XANTHOSINE                          | *   |          |                       |   |               |
| ALLANTOIN                           |   |          |                       |   |               |
| 5'-DEOXYADENOSINE                   | *   |          |                       |   |               |
| METHYLTHIOADENOSINE                 |   |          | *                     |   |               |
| 1-MFTHYLADENOSINE                   |   |          | *                     |   |               |
| AICAR                               |   |          | *                     |   |               |
|                                     | and the second se |          | -                     |   |               |

**Figure 2.** The effect of plastic response and evolutionary adaptation to poor larval diet on abundance of individual metabolites in 3rd instar larvae. For each metabolite, the first column of the heat map represents the estimated effect of diet (poor–standard) on Control larvae; the second column shows the effect of evolutionary regime (Selected–Control) on larvae raised on poor diet. \*q < 0.1 for the corresponding contrast; (\*) marks metabolites with q < 0.1 for the main effect of evolutionary regime but not for Selected–Control contrast on poor diet; "x" between columns indicates regime × diet interaction (q < 0.1, for patterns of those metabolites see Figure 4 and Supplementary Figure S3). Several metabolites show diet effect with log-fold changes greater than ±1 (Supplementary Figure S2); the color scale only varies between -1 and 1 to optimize resolution within this range. For a more traditional heatmap based on clustering of individual populations see Supplementary Figure S4.

factor. However, visual inspection of Figure 2 suggests no consistent relationship between the (presumably ancestral) plastic response of Controls to poor diet and the evolutionary change associated with genetically-based adaptation to it. For example, plasticity induced strong changes in abundance of acyl-carnitines (molecules transporting fatty acids in and out of mitochondria), but no such changes are detected between Selected and Control populations. Amino acids and vitamins tended to be more abundant on poor than standard diet but less abundant in Selected than Control larvae. Among purine compounds a few showed the same direction of change, but for many the evolutionary change tended in the direction opposite to the plastic response.

As another way of exploring the relationship between the evolutionary and plastic responses to poor diet, we plotted the plastic response of the Controls against the difference between Selected on poor and Controls on standard diet (Figure 3A). The latter reflects the difference in metabolome when each set of populations is raised under their evolutionary regime. Under one potential scenario (corresponding to Figure 1B in Ghalambor et al., 2007), the Control plastic response would represent a loss of homeostasis under a novel stress; evolution in Selected populations would then revert these changes, bringing the metabolome of Selected larvae on the poor diet to a similar state as that of Controls on standard diet. If so, most points in Figure 3A should have lain along the X-axis. This is clearly not the case; rather, most points lay along the diagonal and almost all metabolites that showed plastic response to diet have the same sign on X- and Y-axes. Thus, even where evolution and plasticity acted in opposite direction, evolution usually only reduced the consequences of plasticity. Only for a few metabolites was the evolutionary change large enough to reverse the ancestral plastic response.

A statistical assessment of the relationship between plasticity and evolution based on the values plotted in Figure 2 would be potentially biased due to nonindependence of errors; therefore, we based such an assessment on the main effects of diet and regime from the factorial model (see Supplementary Methods). The overlap between metabolites significantly affected by diet and regime (44 metabolites; Figure 1B) was very close to the random expectation of 42.9 (Fisher's exact test, p = .62). We also found no correlation between the estimated main effects of regime (Selected – Control) and diet (poor – standard), whether considering all metabolites (Pearson's r = 0.002, p = .98) or only those significant for both effects (Figure 3B; r = 0.14, p = .36). In this last set, 21 metabolites showed the same sign of plasticity and evolutionary change (i.e., up in Selected and on poor diet or down in Selected and on poor diet) whereas 23 showed opposite signs, not different from random expectation (Fisher's exact test, p = 1). Hence, the plastic response of metabolome to diet did not generally predict the evolutionary response, whether in terms of metabolites affected or the direction of change.

We also found a signature of evolutionary change in the magnitude of plasticity. Even though the estimated effects of diet on metabolite abundance were highly correlated between the Control and Selected regimes, the major axis regression linking them had the slope of 0.89, slightly but significantly less than 1 (Figure 3C; p < .001). In other words, the response of Selected larvae to diet was on average 11% smaller in magnitude than that of Control larvae. This implies that metabolic state of Selected evolved to be somewhat more robust to diet change than that of Controls.

#### BCAA are limiting for development on poor diet

The lower concentration of many amino acids in Selected larvae may seem paradoxical, given that they extract and absorb amino acids from poor diet at a faster rate than Controls (Cavigliasso et al. 2020). Among them, the three branched-chain amino acids (BCAA: valine, leucine, and isoleucine) show a unique pattern of regime  $\times$  diet interaction. All BCAA are depleted in Selected larvae raised on the poor diet, but in Control larvae raised on poor diet they are at least as abundant as in either set of larvae raised on standard diet (Figure 4A–C). (Proline, the fourth amino acid with significant interaction shows a different pattern; Figure 4D). One potential explanation is that maintaining a high level of free BCAA on poor diet is in itself maladaptive, e.g., because of their role in regulation of TOR signaling (Soultoukis & Partridge, 2016;



**Figure 3.** Plasticity versus evolutionary change. (A) Adaptation to poor diet as modifier of ancestral plastic response: the relationship between metabolite abundance in Control and Selected larvae raised on poor diet, both expressed as a deviation from the ancestral state (Control larvae on poor diet). Colored symbols indicate metabolites with significant effect (q < 0.1) of diet on Control larvae ("diet in CTL"), of evolutionary regime on larvae raised on poor diet ("regime and diet"). CTL: Controls; SEL: Selected. std: standard diet. (B) The relationship between plastic and evolutionary responses to diet expressed as the main effects of diet and regime from the factorial model (see Methods). Each symbol corresponds to one metabolite, solid symbols are metabolites significant (q < 0.1) for the effects of both evolutionary regime. (C) The relationship between plastic responses of metabolome of Control and Selected populations. Solid symbols: metabolites significant (q < 0.1) for the main effect of diet; solid line represents the major axis regression fitted to all metabolites, dashed line is the diagonal. Correlations were calculated using all metabolites.



**Figure 4.** Branched-chain amino acids (BCAA) and adaptation to poor diet. Top row: Patterns of abundance (mean  $\pm$  SE) of amino acids that show significant regime × diet interaction in the metabolome data: leucine (A), isoleucine (B), valine (C) (BCAA) and proline (D); n.s., q > 0.1; \*q < 0.1; \*q < 0.05; \*\*q < 0.05; \*\*q < 0.01. Bottom row: Effects of supplementing poor diet with BCAA or a mix of tryptophan, histidine, and lysine (WHK) on egg-to-adult survival probability (E), developmental rate (F), female dry weight (G), and female estimated growth rate (H). Symbols indicate means  $\pm$  SE. N = 3 replicate bottles per population and treatment. CTL: Controls. SEL: Selected. Reg: regime. Treat: treatment. Details of statistics in Supplementary Tables S4–S8.

Juricic et al., 2020) or their correlation with development and physiological disturbance (Tsai et al., 2020). If so, Selected populations might have been selected to reduce their level. Under this scenario, supplementing the poor diet with BCAA would have no or negative effects on larval growth, development, and/or survival. Alternatively, the lower level of amino acids in Selected larvae might reflect their more efficient or immediate use in protein synthesis or metabolism. Under this scenario, the particularly strong reduction of BCAA might indicate that they are limiting for protein synthesis and thus growth and development on the poor diet; hence, supplementing the poor diet with BCAA should lead to improved larval performance.

To test these contrasting predictions, we studied how supplementation of poor diet with BCAA affects larval developmental traits in the Selected and Control populations. To verify that a potential effect is specific to BCAA rather than being mediated by any amino acids, we also investigated the effect of supplementation of the poor diet with a mix of three other essential amino acids, tryptophan, histidine, and lysine (referred to as WHK from one letter abbreviations of those amino acids; Supplementary Figure S5).

Supplementing the poor diet with either BCAA or WHK did not change the egg-to-adult survival probability (Figure 4E; Supplementary Table S5). However, the developmental rate of Selected and Control populations did respond differently to amino acid supplementation (Figure 4F; Supplementary Table S6). This interaction was not driven by differential response to BCAA supplementation: adding BCAA improved developmental rate of both Selected and Controls populations ( $t_{21} = 3.5$ , p = .006 for Controls,  $t_{20} = 4.6$ , p < .001 for Selected) and it did so to a similar degree (interaction contrast (BCAA – Poor)<sub>in SEL</sub> – (BCAA – Poor)<sub>in CTL</sub>:  $t_{20} = 0.7$ , p = .48). By contrast, supplementing poor diet with WHK tended to affect Control and Selected populations differently ((WHK – Poor)<sub>in SEL</sub> – (WHK – Poor)<sub>in CTL</sub>:  $t_{20} = 2.1$ , p = .051), with no detectable change of developmental rate in Controls (WHK – Poor:  $t_{20} = 0.5$ , p = .90) and a negative effect in Selected larvae (Figure 4F; WHK – Poor:  $t_{20} = 3.4$ , p = .008).

We found no effect of amino acid supplementation on adult female dry weight (Figure 4G; Supplementary Table S7), despite the faster development of BCAA-supplemented larvae. Consistent with this, BCAA supplementation improved larval growth rate (Figure 4H; BCAA – Poor:  $t_{20} = 2.8$ , p = .029; Supplementary Table S8). As reported before (Kolss et al., 2009; Erkosar et al., 2017; Cavigliasso et al., 2020), Selected larvae grew faster than Controls but still produced smaller adults because of shortened developmental time; however the effects of amino acid supplementation treatments on female weight and growth rate did not differ between Selected and Control populations (Supplementary Tables S7 and S8).

Thus, BCAA—and not the other three amino acids we tested appeared to be limiting for larval growth and developmental rate on the poor diet, although they did not seem to affect the survival or adult weight.

### Selected larvae catabolize amino acids at a higher rate

In addition to a more immediate use of amino acids in protein synthesis, the apparent depletion of multiple amino acids in Selected larvae could be mediated by increased use of amino acids for energy generation. This process involves deamination and excretion of the resulting nitrogen as urate (uric acid) synthesized in the purine synthesis pathway (Salway, 2018). Consistent with a higher deamination and nitrogen excretion rate, the Selected larvae show an increased concentration of urate and its immediate precursor xanthine, as well as of AICAR, a nucleotide that carries four nitrogen atoms from deamination into purine synthesis (Figure 2, Figure 5A and B). As independent test of this hypothesis, we quantified the relative content of <sup>15</sup>N isotope ( $\Delta$ <sup>15</sup>N) at the end of larval development (at the prepupal stage) and in freshly emerged adult flies—a higher rate of deamination leads to a higher <sup>15</sup>N/<sup>14</sup>N ratio. In agreement with our hypothesis, at both stages and irrespective of diet the Selected populations had a higher  $\Delta$ <sup>15</sup>N value than Controls (Figure 5C;  $F_{1,11} = 18.9$ , p = .001; Supplementary Table S9). Furthermore, individuals of both sets of populations accumulated more <sup>15</sup>N when raised on standard versus poor diet (Figure 5C,  $F_{1,12} = 88.4$ , p < .001). Thus, consistent with the predictions



**Figure 5.** Plasticity and evolution of amino acid deamination in response to poor diet. (A) Simplified purine/uric acid synthesis pathway (created in Lucidchart, www.lucidchart.com). Metabolites are colored according to their estimated fold change (Selected – Control) on poor diet; \*q < 0.1. XMP was not reliably quantified. (B) Relative concentration of urate in Selected and Control larvae raised on poor and standard diet. (C) Isotope content after feeding on poor or standard diet enriched in <sup>15</sup>N.  $\Delta$ <sup>15</sup>N (‰ N<sub>2</sub>-Air) refers to the difference between  $\delta$ <sup>15</sup>N of prepupae or adults and  $\delta$ <sup>15</sup>N of the larval food. Symbols indicate means ± SE. Details of statistics in Supplementary Table S9. N = 2 samples of 20–30 individuals per population, diet, and life stage.

derived from the metabolome analysis, larvae confronted with a nutrient-poor diet reduced the rate at which they catabolize amino acids, but the Selected larvae catabolized a greater fraction of assimilated amino acids than Controls.

## **Discussion** Plasticity, evolutionary adaptation, and metabolic efficiency

As expected, nutrient restriction imposed by the poor diet resulted in large changes in metabolite abundance. What was unexpected is that more metabolites increased than decreased in abundance in larvae raised on the poor diet relative to those raised on standard diet. The metabolome data are normalized to the sample protein content. Assuming that protein content is a measure of structural size of the larva, differences in normalized metabolite abundance should thus reflect differences in their relative concentration in the cells and hemolymph. Therefore, the greater abundance of core metabolites such as group B vitamins, nucleotide cofactors (FAD+, NAD+, NADP+, ATP), and multiple amino acids in larvae raised on the poor diet appears paradoxical: compared to the standard diet, the poor diet contains 25% of all dietary sources of these metabolites or their precursors. To our knowledge, no other study has reported consequences of larval dietary restriction for metabolome, but two studies report an increase in abundance of some amino acids in response to dietary restriction at an adult stage (Supplementary Table S5 in Laye et al., 2015, Supplementary Table A in Jin et al., 2020). Thus, dietary availability of protein does not seem to lead to generally lower concentrations of amino acids in the tissues.

These changes in metabolite abundance obviously do not mean that the larvae are somehow better off on the poor diet. Possibly, the accumulation of some amino acids and most other metabolic changes in poor-diet raised larvae, at least those from the Control populations, reflect a stress-induced perturbation of homeostasis that hampers the optimal use of even the small quantity of nutrients the larvae manage to extract. Alternatively, the metabolic responses to poor diet might mostly reflect adaptive phenotypic plasticity in metabolism that alleviates the consequences of undernutrition. Even though the Control populations were not exposed to the poor diet during the experimental evolution, such adaptive plastic responses might have evolved in wild ancestors under variable nutrient availability in nature.

It has been argued that one could ascertain the adaptive versus nonadaptive (or maladaptive) nature of plasticity in response to a novel environmental stress by examining genetically-based evolutionary change driven by same stress. If the sign of plastic response and evolutionary change in a phenotype is the same ("synergistic"), it would suggest that the plastic response is adaptive; if the signs are opposite ("antagonistic"), the plastic response would be maladaptive (Yampolsky et al., 2012; Ghalambor et al., 2015; Huang & Agrawal, 2016; Josephs et al., 2021). In this light, the studies just cited explored the correlation between phenotypically plastic and evolutionary changes in gene expression patterns, the first three finding a negative correlation and the last one a positive one. Here, we found no overall correlation between plastic and evolutionary responses of metabolite abundance to diet; for metabolites that showed a plastic response, evolution was as likely to act in the same as in the opposite direction to plasticity. However, nearly all metabolites that increased (decreased) in abundance in the Control larvae in response to the poor diet remained more (less) abundant in the Selected larvae on poor diet than in Control larvae on standard diet. In other words, even if for many metabolites the evolutionary change in metabolite abundance was antagonistic to plasticity, in most cases it was too small to completely offset or to reverse the plastic response. This can be interpreted in two ways. Possibly, in case of these metabolites the plastic response was maladaptive and thus opposed by selection in the Selected populations, but there was not enough time and/or genetic variability for evolution to bring the phenotype to the ancestral state. Alternatively, the initial plastic response may have been adaptive in terms of general direction, but imprecise, overshooting the optimal level. Under the latter scenario, the phenotypic plasticity of metabolism may at the same time be adaptive (compared to no response) and induce some loss of metabolic homeostasis. If so, the evolutionary changes in either direction would represent fine-tuning of the initial plastic response.

We tend to favor the latter interpretation. A maladaptive plastic response would bring the phenotype farther away from the optimum than an adaptive response. Assuming that the strength of directional selection is greater farther away from the optimum, selection to revert maladaptive plastic responses should be stronger than selection to refine adaptive responses. Yet, we see no general tendency for the evolutionary changes that are antagonistic to the initial plasticity to be larger in magnitude than those that are synergistic with it, suggesting that plastic responses opposed by evolutionary change are not more maladaptive than those where evolution acts in the same direction as plasticity. While this is a speculation, this argument illustrates that it may be too simplistic to interpret plastic responses as adaptive or not solely based on the sign of correlation between plasticity and evolutionary change driven by the same environmental factor.

While the plastic response to poor diet led to more increases than decreases in metabolite abundance, of the metabolites affected by evolutionary adaptation to the poor diet nearly 3/4 decreased. This trend held regardless of the presence and direction of the plastic response of these metabolites. Many of these metabolites are building blocks of macromolecules (amino acids, nucleotides, and nucleosides), precursors of cell membrane lipids, and intermediates in energy metabolism. Some of them can be derived from dietary sugars and so their lower levels might be explained by the lower assimilation of carbohydrates by Selected than Control larvae (Cavigliasso et al., 2020). However, the Selected larvae assimilate amino acids from poor diet at a higher rate than Controls (Cavigliasso et al., 2020). Thus, rather than reflecting a supply bottleneck, the lower abundance of amino acids in Selected larvae may result from their more efficient use. The metabolic cofactors NAD+, NADP+, and FAD+ and their vitamin precursor were also lower in Selected than Control larvae on the poor diet but still above Control larvae on standard diet; thus, they were unlikely to be deficient to the degree that would limit metabolic flux. While we do not have any direct measure of metabolic fluxes, the fact that on poor diet Selected larvae grow faster than Controls implies that they must achieve a higher rate of protein, nucleic acid, and membrane lipid synthesis, as well as being able to provide for the energy cost of the faster growth. Indeed, despite lower abundance of many energy metabolism intermediates and electron-carrying cofactors, Selected larvae do not have a lower level of ATP than Controls. Our results thus support the notion that experimental adaptation to larval undernutrition has been mediated in part by more efficient metabolism that reduced the concentration of cellular "building blocks" and metabolic intermediates while supporting faster growth in the face of dietary nutrient limitation. They also show that the abundance of free amino acids does not in general reflect the nutritional condition of an animal and its growth potential.

# Trade-off in nutrient acquisition leads to "waste" of amino acids for energy generation

Among the minority of metabolites whose abundance on poor diet was higher in Selected than Control larvae, most are amino acid derivatives. While some have known physiological functions in the nervous system (GABA, N-acetylaspartate) or muscles (creatine), many are products of amino acid degradation with no known physiological function, including the main nitrogenous excretion compound urate (uric acid). Consistent with the pattern of urate abundance, our <sup>15</sup>N accumulation experiment demonstrated that, while Control larvae nearly completely cease deaminating amino acids on the poor diet, in the Selected larvae the rate of deamination on poor diet was restored to about half of that observed on standard diet. This implies that the ancestral plastic response to nutrient shortage was to minimize "wasting" of amino acids, but that adaptation to the poor diet favored a moderate use amino acids for energy generation.

This might be explained by the Selected populations having evolved the ability to assimilate amino acids from the poor diet at a higher rate (Cavigliasso et al., 2020). However, Selected larvae still grow considerably more slowly on the poor than standard diet. Arguably, using these amino acids for protein synthesis instead of deaminating them might enable an even faster growth and development. Faster growth requires increased supply of ATP; just protein synthesis consumes about 5 ATP molecules per attached amino acid (Marchingo & Cantrell, 2022). Yet, the Selected larvae have become less efficient in extracting carbohydrates from the poor diet (Cavigliasso et al., 2020). While some of this reduction in carbohydrate acquisition may be compensated by lower accumulation of fat stores (Cavigliasso et al., 2020), it may also lead to a gap between the energetic needs and the supply of carbohydrate energy sources. In other words, it is not just that the Selected larvae and pupae can afford to "waste" some amino acids for energy; they may be compelled to do so by an evolutionary trade-off in acquisition of amino acids versus carbohydrates from the diet.

# A special role for branched chain amino acids in adaptation to undernutrition?

The three branched-chain amino acids—leucine, isoleucine, and valine—showed a unique pattern of interaction, remaining similar or increasing in response to poor diet in Control larvae but strongly declining in Selected larvae (Figure 4A–C). This was mediated entirely by a change in the plastic response to poor diet; on standard diet Selected and Control larvae showed nearly identical levels of all three BCAA. This suggests that selection driven by undernutrition favored a mechanism that resulted in poor diet-specific reduction in their abundance. This could have been because low levels of BCAA in itself promote growth and development on poor diet, or because an increase in the rate of protein synthesis evolved by the Selected populations depletes these amino acids to a greater degree than most others.

Our results support the latter scenario. We showed that supplementing poor diet with BCAA improves growth of both Selected and Control larvae whereas adding three other essential amino acids does not, suggesting that one or more BCAA are limiting for growth. Leucine is the most abundant amino acid in the fly proteome (9.4% of predicted amino acids) with valine (6.2%) and isoleucine (5.2%) being in the second and sixth by abundance among essential amino acids, respectively (Piper et al., 2017). More importantly, while the latter two BCAA form a similar percentage of amino acids in yeast, leucine content in yeast protein is only about 7.5% (Piper et al., 2017). Thus, being in short supply proportionally to demand, leucine would be depleted more than other essential amino acids if the rate of protein synthesis increased, and thus might become the limiting amino acid. However, Piper et al. (2017) identified methionine as even more limiting; yet, in our study methionine is the only amino acid that tended to increase in Selected larvae relative to Control. Furthermore, the mismatch of amino acid composition in yeast and Drosophila proteins would not predict isoleucine and valine showing a similar decrease as a result of a higher protein synthesis rate. Thus, our results do not allow us to completely exclude the possibility that the depletion of BCAA in Selected larvae on poor diet is (also) mediated by reduced absorption from the diet and/or enhanced degradation. Deamination of all three BCAA is catalyzed by the same enzyme (Blumrich et al., 2021) while their intestinal absorption, at least in mammals, involves the same transmembrane transporters (Bröer & Fairweather, 2018), which would explain parallel changes in abundance of all BCAA.

BCAA, and in particular leucine, play an important signaling role by stimulating the TOR pathway, which regulates protein synthesis (Antikainen et al., 2017). We have thus speculated that a reduction in the level of BCAA under nutrient shortage may have been favored by selection to avoid a situation where TOR signaling promotes initiation of more protein synthesis processes than the cells have amino acids to complete. However, if so, BCAA supplementation should have been harmful, which was not the case. Nonetheless, irrespective of the mechanism, the depletion of BCAA would have consequences for nutrient sensing, possibly requiring compensatory changes in growth regulation. This is consistent with changes in insulin signaling and the expression of FOXO targets we reported before (Erkosar et al., 2017; Kawecki et al. 2021). This interpretation of our results opens the question of potential evolutionary tradeoffs in nutritional adaptation that may result from the same compounds playing a role as regulatory molecules and building blocks of macromolecules or intermediates in energy metabolism.

### Supplementary material

Supplementary material is available online at Evolution Letters (https://academic.oup.com/evlett/qrac018).

### Data availability

The metabolite abundance data, the data of larvae raised on BCAA or WHK or poor diet and the isotope content data are provided in supplemental data, in files named "Raw\_data\_metabolomics," "Raw\_data\_BCAA and WHK supplementation," "Raw\_data\_15N enrichment." R script for statistical analysis is provided in Supplemental data.

### Author contributions

F.C. and T.J.K. designed the research, F.C. and L.S. performed all Drosophila experiments and collected samples, J.E.S measured  $\delta^{15}$ N, H.G.-A. and J.I. performed and oversaw metabolite quantification, F.C., H.G.-A., and T.J.K. analyzed the data. F.C. and T.J.K. wrote the article, with contributions by all authors.

### Funding

This work has been supported by the Swiss National Science Foundation (grants number 31003A\_162732 and 310030\_184791) and research funding from the University of Lausanne.

Conflict of interest: The authors declare no conflict of interest.

## Acknowledgments

We thank Nora Corthésy for help with the experiments and Berra Erkosar for her advice.

### References

- Antikainen, H., Driscoll, M., Haspel, G., & Dobrowolski, R. (2017). TOR-mediated regulation of metabolism in aging. Aging Cell, 16(6), 1219–1233. https://doi.org/10.1111/acel.12689
- Avanesov, A. S., Ma, S., Pierce, K. A., Yim, S. H., Lee, B. C., Clish, C. B., & Gladyshev, V. N. (2014). Age- and diet-associated metabolome remodeling characterizes the aging process driven by damage accumulation. *eLife*, **3**, e02077. https://doi.org/10.7554/eLife.02077
- Behmer, S. T. (2009). Insect herbivore nutrient regulation. Annual Review of Entomology, 54, 165–187. https://doi.org/10.1146/ annurev.ento.54.110807.090537
- Blumrich, A., Vogler, G., Dresen, S., Diop, S. B., Jaeger, C., Leberer, S., Grune, J., Wirth, E. K., Hoeft, B., Renko, K., Foryst-Ludwig, A., Spranger, J., Sigrist, S., Bodmer, R., & Kintscher, U. (2021). Fatbody brummer lipase determines survival and cardiac function during starvation in *Drosophila melanogaster*. iScience, 24(4), 102288. https://doi.org/10.1016/j.isci.2021.102288
- Boggs, C. L. (2009). Understanding insect life histories and senescence through a resource allocation lens. Functional Ecology, 23(1), 27–37. https://doi.org/10.1111/j.1365-2435.2009.01527.x
- Bröer, S., & Fairweather, S. J. (2018). Amino acid transport across the mammalian intestine. Comprehensive Physiology, 9(1), 343–373. https://doi.org/10.1002/cphy.c170041
- Carlsson, H., Ivimey-Cook, E., Duxbury, E. M. L., Edden, N., Sales, K., & Maklakov, A. A. (2021). Ageing as "early-life inertia": Disentangling life-history trade-offs along a lifetime of an individual. Evolution Letters, 5(5), 551–564. https://doi.org/10.1002/ evl3.254
- Cavigliasso, F., Dupuis, C., Savary, L., Spangenberg, J. E., & Kawecki, T. J. (2020). Experimental evolution of post-ingestive nutritional compensation in response to a nutrient-poor diet. *Proceedings* of the Royal Society B, **287**, 20202684. https://doi.org/10.1098/ rspb.2020.2684
- de Magalhaes, J. P. (2012). Programmatic features of aging originating in development: Aging mechanisms beyond molecular damage? FASEB Journal, 26(12), 4821–4826. https://doi.org/10.1096/fj.12-210872
- Erkosar, B., Kolly, S., van der Meer, J. R., & Kawecki, T. J. (2017). Adaptation to chronic nutritional stress leads to reduced dependence on microbiota in *Drosophila melanogaster*. *mBio*, **8**, e01496-17. https://doi.org/10.1128/mBio.01496-17
- Gannes, L. Z., O'Brien, D.M., & Del Rio, C. M. (1997). Stable isotopes in animal ecology: Assumptions, caveats, and a call for more laboratory experiments. *Ecology*, **78**, 1271–1276.
- Ghalambor, C. K., Hoke, K. L., Ruell, E. W., Fischer, E. K., Reznick, D. N., & Hughes, K. A. (2015). Non-adaptive plasticity potentiates rapid adaptive evolution of gene expression in nature. *Nature*, 525(7569), 372–375. https://doi.org/10.1038/nature15256
- Ghalambor, C. K., McKay, J. K., Carroll, S. P., & Reznick, D. N. (2007). Adaptive versus non-adaptive phenotypic plasticity and the potential for contemporary adaptation in new environments. *Functional Ecology*, **21**(3), 394–407. https://doi. org/10.1111/j.1365-2435.2007.01283.x
- Holtof, M., Lenaerts, C., Cullen, D., & Vanden Broeck, J. (2019). Extracellular nutrient digestion and absorption in the insect gut. Cell and Tissue Research, 377(3), 397–414. https://doi.org/10.1007/ s00441-019-03031-9

- Huang, Y., & Agrawal, A. F. (2016). Experimental evolution of gene expression and plasticity in alternative selective regimes. *PLoS Genetics*, **12**(9), e1006336. https://doi.org/10.1371/journal. pgen.1006336
- Jin, K., Wilson, K. A., Beck, J. N., Nelson, C. S., Brownridge, G. W., 3rd, Harrison, B. R., Djukovic, D., Raftery, D., Brem, R. B., Yu, S., Drton, M., Shojaie, A., Kapahi, P., & Promislow, D. (2020). Genetic and metabolomic architecture of variation in diet restriction-mediated lifespan extension in Drosophila. PLoS Genetics, 16(7): e1008835. https://doi.org/10.1371/journal.pgen.1008835
- Josephs, E. B., Van Etten, M. L., Harkess, A., Platts, A., & Baucom, R. S. (2021). Adaptive and maladaptive expression plasticity underlying herbicide resistance in an agricultural weed. *Evolution Letters*, 5(4), 432–440. https://doi.org/10.1002/evl3.241
- Juricic, P., Gronke, S., & Partridge, L. (2020). Branched-chain amino acids have equivalent effects to other essential amino acids on lifespan and aging-related traits in Drosophila. The Journals of Gerontology. Series A: Biological Sciences and Medical Sciences, 75(1), 24–31. https://doi.org/10.1093/gerona/glz080
- Kawecki, T. J., Erkosar, B., Dupuis, C., Hollis, B., Stillwell, R. C., & Kapun, M. (2021). The genomic architecture of adaptation to larval malnutrition points to a trade-off with adult starvation resistance in Drosophila. Molecular Biology and Evolution, **38**(7), 2732–2749. https://doi.org/10.1093/molbev/msab061
- Kolss, M., Vijendravarma, R. K., Schwaller, G., & Kawecki, T. J. (2009). Life-history consequences of adaptation to larval nutritional stress in Drosophila. Evolution, 63(9), 2389–2401. https://doi. org/10.1111/j.1558-5646.2009.00718.x
- Lande, R. (2009). Adaptation to an extraordinary environment by evolution of phenotypic plasticity and genetic assimilation. *Journal of Evolutionary Biology*, **22**(7), 1435–1446. https://doi. org/10.1111/j.1420-9101.2009.01754.x
- Laye, M. J., Tran, V., Jones, D. P., Kapahi, P., & Promislow, D. E. (2015). The effects of age and dietary restriction on the tissue-specific metabolome of Drosophila. Aging Cell, 14(5), 797–808. https://doi. org/10.1111/acel.12358
- Manière, G., Alves, G., Berthelot-Grosjean, M., & Grosjean, Y. (2020). Growth regulation by amino acid transporters in Drosophila larvae. Cellular and Molecular Life Sciences, 77(21), 4289–4297. https:// doi.org/10.1007/s00018-020-03535-6
- Marchingo, J. M., & Cantrell, D. A. (2022). Protein synthesis, degradation, and energy metabolism in T cell immunity. Cellular and Molecular Immunology, 19(3), 303–315. https://doi.org/10.1038/ s41423-021-00792-8
- Minagawa, M., & Wada, E. (1984). Stepwise enrichment of 15N along food chains: Further evidence and the relation between δ15N and animal age. Geochimica et Cosmochimica Acta, 48(5), 1135– 1140. https://doi.org/10.1016/0016-7037(84)90204-7
- Paenke, I., Sendhoff, B., & Kawecki, T. J. (2007). Influence of plasticity and learning on evolution under directional selection. American Naturalist, 170, E47–E58.
- Peterson, B. J., & Fry, B. (1987). Stable isotopes in ecosystem studies. Annual Review of Ecology and Systematics, 18(1), 293–320. https:// doi.org/10.1146/annurev.es.18.110187.001453
- Pigliucci, M., & Murren, C. J. (2003). Genetic assimilation and a possible evolutionary paradox: Can macroevolution sometimes be so fast as to pass us by? Evolution, 57(7), 1455–1464. https://doi. org/10.1111/j.0014-3820.2003.tb00354.x
- Piper, M. D. W., Soultoukis, G. A., Blanc, E., Mesaros, A., Herbert, S. L., Juricic, P., He, X., Atanassov, I., Salmonowicz, H., Yang, M., Simpson, S. J., Ribeiro, C., & Partridge, L. (2017). Matching dietary amino acid balance to the in silico-translated

exome optimizes growth and reproduction without cost to lifespan. Cell Metabolism, **25**(5), 1206. https://doi.org/10.1016/j. cmet.2017.04.020

- Prentice, A. M. (2005). Starvation in humans: Evolutionary background and contemporary implications. Mechanisms of Ageing and Development, **126**(9), 976–981. https://doi.org/10.1016/j. mad.2005.03.018
- Rodrigues, M. A., Martins, N. E., Balance, L. F., Broom, L. N., Dias,
  A. J., Fernandes, A. S., Rodrigues, F., Sucena, E., & Mirth,
  C. K. (2015). Drosophila melanogaster larvae make nutritional choices that minimize developmental time. Journal of Insect Physiology, 81, 69–80. https://doi.org/10.1016/j. jinsphys.2015.07.002
- Roseboom, T., de Rooij, S., & Painter, R. (2006). The Dutch famine and its long-term consequences for adult health. Early Human Development, 82(8), 485–491. https://doi.org/10.1016/j. earlhumdev.2006.07.001
- Salway, J. G. (2018). The Krebs uric acid cycle: A forgotten Krebs cycle. Trends in Biochemical Sciences, **43**(11), 847–849. https://doi. org/10.1016/j.tibs.2018.04.012
- Scheiner, S. M. (1993). Genetics and evolution of phenotypic plasticity. Annual Review of Ecology and Systematics, **24**(1), 35–68. https://doi.org/10.1146/annurev.es.24.110193.000343
- Simpson, S. J., Raubenheimer, D., Simpson, S. J., & Raubenheimer, D. (2012). Mechanisms of nutritional regulation. In *The nature of nutrition: A unifying framework from animal adaptation to human obesity*. Princeton University Press, pp. 35–56.
- Singh, P. P., Demmitt, B. A., Nath, R. D., & Brunet, A. (2019). The genetics of aging: A vertebrate perspective. *Cell*, **177**(1), 200–220. https://doi.org/10.1016/j.cell.2019.02.038

- Soultoukis, G. A., & Partridge, L. (2016). Dietary protein, metabolism and aging. Annual Review of Biochemistry, 85, 5–34.
- Tatar, M., Bartke, A., & Antebi, A. (2003). The endocrine regulation of aging by insulin-like signals. *Science*, **299**(5611), 1346–1351. https://doi.org/10.1126/science.1081447
- Tsai, H. Y., Wu, S. C., Li, J. C., Chen, Y. M., Chan, C. C., & Chen, C. H. (2020). Loss of the Drosophila branched-chain alpha-ketoacid dehydrogenase complex results in neuronal dysfunction. Disease Models & Mechanisms, 13, dmm044750. https://doi. org/10.1242%2Fdmm.044750
- Vijendravarma, R. K., Narasimha, S., & Kawecki, T. J. (2011). Plastic and evolutionary responses of cell size and number to larval malnutrition in Drosophila melanogaster. Journal of Evolutionary Biology, 24(4), 897–903. https://doi. org/10.1111/j.1420-9101.2010.02225.x
- Vijendravarma, R. K., Narasimha, S., & Kawecki, T. J. (2012). Chronic malnutrition favours smaller critical size for metamorphosis initiation in Drosophila melanogaster. Journal of Evolutionary Biology, 25(2), 288–292. https://doi. org/10.1111/j.1420-9101.2011.02419.x
- Wu, G., Bazer, F. W., Dai, Z., Li, D., Wang, J., & Wu, Z. (2014). Amino acid nutrition in animals: Protein synthesis and beyond. Annual Review of Animal Biosciences, 2, 387–417. https://doi.org/10.1146/ annurev-animal-022513-114113
- Yampolsky, L. Y., Glazko, G. V., & Fry, J. D. (2012). Evolution of gene expression and expression plasticity in longterm experimental populations of Drosophila melanogaster maintained under constant and variable ethanol stress. Molecular Ecology, **21**(17), 4287–4299. https://doi. org/10.1111/j.1365-294X.2012.05697.x